## Author's response to reviews

**Title:**An Empirical Study to Determine Factors That Motivate and Limit The Implementation Of ICT In Healthcare Environments

## Authors:

Raj Gururajan (<u>gururaja@usq.edu.au</u>) Abdul Hafeez-Baig (<u>abdulhb@usq.edu.au</u>)

## Version:8Date:23 June 2014

## Author's response to reviews:

Already provided